Author's response to reviews

Title: Comparative Efficacy of Six Therapies for Hypopharyngeal and Laryngeal Neoplasms: a Network Meta-analysis

Authors:

Juan Che (zyyanzhenguo@126.com)
Yanlin Wang (zyysht0707@126.com)
Xiaolin Zhang (zyyyyyjm@126.com)
Jun Chen (zhry960034@bzmc.edu)

Version: 4 Date: 19 Feb 2019

Author's response to reviews:

Editor Comments:

1 - Please provide an institutional email address on the title page and in the submission data for corresponding author Jun Chen; we were unable to verify the currently listed email for this author.

RE: Dear Editor, an institutional email address for corresponding author Jun Chen has been provided on the title page and in the submission data.

3 - Please use (unique) initials to distinguish between authors in the Authors contributions.

RE: Dear Editor, we have revised the Authors contributions as you required.

Authors' contributions: JC(Che) made substantial contributions to conception and design. YW made acquisition of data. XZ and JC(Chen) made analysis and interpretation of data. JC(Che) have been involved in drafting the manuscript. YW, XZ and JC(Chen) have been involved in revising it critically for important intellectual content. All authors have given final approval of the version to be published. All authors agreed to be accountable for all aspects of the work.

4 - Please remove the files in the file inventory that are not required for publication such as Manuscript-with track.docx
RE: Dear Editor, the files that are not required for publication have been removed.

5 - At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

RE: Dear Editor, we have uploaded our manuscript as a single, final, clean version, and all relevant tables and figures are also clean versions. Figures have been uploaded as separate files. All the requirements have been followed.